Author’s response to reviews

Title: The Effect of Neoadjuvant Chemotherapy and Chemoradiotherapy on Exercise Capacity and Outcome Following Upper Gastrointestinal Cancer Surgery: An Observational Cohort Study

Authors:

Malcolm West (m.west@soton.ac.uk)
Lisa Loughney (lisa.loughney@uhs.nhs.uk)
Gareth Ambler (g.ambler@ucl.ac.uk)
Borislav Dimitrov (b.dimitrov@soton.ac.uk)
Jamie Kelly (jamie.kelly@uhs.nhs.uk)
Michael Mythen (m.mythen@ucl.ac.uk)
Richard Sturgess (richard.sturgess@aintree.nhs.uk)
Peter Calverley (pmacal@liverpool.ac.uk)
Adrian Kendrick (adrian.kendrick@uhbristol.ac.uk)
Michael Grocott (mike.grocott@soton.ac.uk)
Sandy Jack (s.jack@soton.ac.uk)

Version: 2 Date: 15 Jun 2016

Author’s response to reviews:

We have added a separate 'Declarations' section which is highlighted at the bottom of the manuscript. This was the only modification that we made.